Suppressed Daytime Convection over the Amazon River

- 2 M. Wu^{1,2}, J.-E. Lee^{1,2}, D. Wang³, and M. Salameh²
- 3 ¹Department of Earth, Environmental and Planetary Sciences, Brown University, Providence, RI,
- 4 USA.

1

- ²Institute at Brown for Environment and Society, Brown University, Providence, RI, USA.
- 6 ³Department of Environmental and Climate Sciences, Brookhaven National Laboratory, Upton,
- 7 NY, USA.
- 8 Corresponding author: Mengxi Wu (mengxi wu@brown.edu)

9 **Key Points:**

- Daytime rainfall is suppressed over the Amazon River compared with surrounding forest.
- The reduction of daytime rainfall has a significantly negative linear relationship with the Laplacian of surface temperature.
- Daytime rain over the river needs moisture convergence above the surface, while above the forest most moisture converges near the surface.

Abstract

We investigated the interaction between surface conditions and precipitating convection by comparing the Amazon River against the surrounding forest. Despite similar synoptic conditions within a few tens of kilometers, the river surface is substantially cooler than the surrounding forest in the day and warmer at night. We analyzed twenty years of high-resolution satellite precipitation data and confirmed previous findings of daytime rainfall reduction over the river. This reduction is strongest during the dry-to-wet transition season. In addition, the reduction of each tributary is significantly correlated with a positive Laplacian of surface temperature, which causes thermally-driven surface divergence and suppresses local convection. Additionally, nighttime rainfall is enhanced over tributaries near the coast during the wet season. The local rainfall anomalies associated with the river is then simulated by a regional climate model. Above the river, moisture diverges near the surface and converges above the surface before the daytime rainfall, partially driven by the horizontal gradient of humidity. Unlike the river, moisture convergence within the boundary layer is more important for the rainfall above the forest region. Our studies suggest that strong thermal contrast can be important in deriving heterogeneous convection in moist tropical regions.

Plain Language Summary

To understand how surface types and surface conditions influence precipitation and cloud processes, we compared rainfall characteristics between the Amazon River and its surrounding rainforest using satellite observations and a regional climate model. With similar large-scale meteorological conditions, a much cooler river surface is shown to reduce daytime rainfall relative to the forest. By contrast, the warmer river surface at night contributes to higher nighttime rainfall in some tributaries near the coast. The climate model further suggests that daytime rain over the river relies on moisture source above the surface, whereas above the rainforest most moisture converges near the surface.

1 Introduction

Along with its vast tropical rainforest, Amazonia is a key area to study atmospheric convection for its unique convective characteristics. There are abundant maritime-like congestus clouds (Wall et al., 2013; D. Wang et al., 2018), intermediate intensity of thunderstorms (Zipser

et al., 2006), more mature mesoscale convective systems (MCSs) relative to other continents (Bang & Zipser, 2016; D. Wang et al., 2019, 2020), and very low aerosol concentration during the wet season (Andreae et al., 2004; Williams et al., 2002).

Atmospheric moisture is one of the important factors that determine the convective characteristics. Column-integrated water vapor (Bretherton et al., 2004; Neelin et al., 2009; Peters & Neelin, 2006), or more specifically, lower-tropospheric water vapor, has a robust relationship with deep convection either over the ocean (Holloway & Neelin, 2009; Schiro & Neelin, 2019) or over the land (Schiro & Neelin, 2019; Zhang & Klein, 2010; Zhuang et al., 2017). Modelling studies suggest that this relationship is achieved through entrainment of convective updrafts in the lower troposphere (Kuo et al., 2017), which substantially reduces the buoyancy of rising air if the lower troposphere is dry (Schiro et al., 2018).

With different entrainment strength (Lucas et al., 1994; Takahashi et al., 2017), the oceanic convection is mostly associated with moisture in the lower free troposphere (Holloway & Neelin, 2009; Schiro & Neelin, 2019; Wu & Lee, 2019), while daytime continental convection in Amazonia is additionally linked to the boundary layer moisture considering both precipitation intensity (Schiro & Neelin, 2019) and cloud vertical structures (Wu & Lee, 2019). Furthermore, the importance of boundary layer moisture may depend on seasons, as the differences in preconditioning between shallow and deep convection shows a larger signal in boundary layer moisture during the dry-to-wet transition (Zhuang et al., 2017). Climate model studies have shown that boundary layer humidity also lowers the convective inhibition (CIN), which modify the spatial and temporal patterns of precipitation (Itterly et al., 2018).

In addition to atmospheric moisture, surface heterogeneity is also crucial to convection over land. Warm surface temperature can induce boundary layer deepening and consequent convective initiation (Gentine et al., 2013). On a small spatial scale, heterogeneous surface temperature and sensible heat flux can drive shallow circulations and induce convergence over the warm surface (Hohenegger & Stevens, 2018; Lindzen & Nigam, 1987; Taylor & Ellis, 2006). Studies (e.g., Taylor et al., 2012) have shown that thermally driven convergence may be more prominent in semi-arid regions and determines a negative soil moisture feedback for afternoon precipitation. In Amazonia, by contrast, researchers (Gentine et al., 2019; J. F. Wang et al., 2009) claim that the temperature effect is important only for shallow convection instead of deep

precipitating convection when comparing unperturbed rainforest against neighboring deforested pastures.

A possible explanation for the weak impact of surface temperature gradient between the Amazonian forest and the pasture on precipitation may be the small magnitude of temperature contrast. For instance, the river breeze circulation driven by a much larger surface temperature gradient during the daytime is reported to cause rainfall suppression over the Amazon River. Station rain gauges, satellites and ground-based radars all observe a noticeable reduction of rainfall over the river during the day and an increase at night at some locations (Cohen et al., 2014; Fitzjarrald et al., 2008; Paiva et al., 2011). This diurnal pattern of rainfall differences is consistent with the diurnal reversal of surface temperature gradient near the river and the river breeze circulation (Dias et al., 2004).

The distinct surface type within a short distance near the Amazon River is beneficial for understanding how temperature influences precipitating convection. Particularly, it provides a desirable condition for isolating the influence of the surface, as its proximity to the surrounding forest imply similar synoptic conditions. In addition, the deep roots of the rainforest lead to only moderate water stress in the dry season (Nepstad et al., 1999; Lee et al., 2005), and the surface latent heat flux and humidity are fairly high (Gentine et al., 2019; Wu & Lee, 2019). Thus, the differences between the river and the forest will predominantly be surface temperature, which induces a local river breeze circulation (Dias et al., 2004).

In this study, we investigate the differences in rainfall between the Amazon River and the surrounding forest to understand the temperature effects on precipitation. Using the Climate Prediction Center morphing method (CMORPH), a high-resolution satellite precipitation product (Joyce et al., 2004), the seasonality, the diurnal cycle of the river influence over convection and the differences among tributaries are discussed. We focus on the temperature effect in determining the temporal and spatial variability of rainfall. Further analysis of horizontal moisture convergence and its vertical structure are conducted with the Weather Research and Forecasting (WRF) model, which helps establish the mechanistic understanding of the temperature effect on convection in this region.

The remaining parts of the paper are organized as follows. Datasets and methods are described in Section 2. The diurnal cycle, seasonal cycle, and regional difference of the river-

associated rainfall anomalies are analyzed in Section 3 using the CMORPH precipitation product (Joyce et al., 2004). Contributions from thermally-driven circulation is discussed in this section. Afterwards, we continue to discuss the vertical profile of moisture convergence simulated by the WRF model in Section 4. Distinct profiles over the river and the surrounding forest provide another aspect to understand how the river breeze circulation affect precipitating convection. The paper is concluded in Section 5 with a synthesis of our results and some implications.

2 Materials and Methods

2.1 Area of interest

The area of interest is located in the lower part of the Amazon River near the mouth (Figure S1a). The northwest corner of the domain is wet all year round, while most of the rest area has a rainy season broadly starting in October - December and ending in May – July (Marengo et al., 2001). The coastal regions are strongly influenced by sea breeze and squall lines, the effect of which decreases towards the inland region (Greco et al., 1994). Except for the southern extension of the Guiana Highlands in the northwest corner, the surface elevations within the domain are below 500 m and broadly rise from the west to the east (Figure S1b).

We identified seven major tributaries of the river within the domain: The Negro River, the Solimões River (the upper stretch of the Amazon, including the ends of the Japurá River and the Juruá River in our domain), the Amazon River (the lower stretch of the Amazon), the Madeira River, the Tapajós River, the Branco River and the Purus River (Figure S1a). The first five tributaries are relatively wider while the last two are fairly narrow. In addition, the Usina Hidrelétrica de Balbina (UHE Balbina; the Balbina Hydroelectric Dam) is also clearly identified from the land water mask.

2.2 Datasets

2.2.1 Climate Prediction Center morphing method (CMORPH)

CMORPH is a high-resolution (8 km in space and 30 min in time) precipitation product retrieved from multiple polar-orbiting passive microwave sensors (Joyce et al., 2004). Compared with the 0.25°×0.25° 3-hourly Tropical Rainfall Measuring Mission (TRMM) precipitation product used in Paiva et al. (2011) and the 0.25°×0.25° CMORPH product used in Fitzjarrald et

al. (2008), 8-km CMORPH has an improved ability to detect locally strong precipitation gradients over the Amazon River as well as the diurnal cycle. Although CMORPH is reported to overestimate rainfall near large water bodies in a light rain regime (Tian & Peters-Lidard, 2007), Fitzjarrald et al. (2008) showed a good agreement between CMORPH and station measurements near the Amazon-Tapajós confluence. We analyzed the CMORPH precipitation during 1998-2017.

2.2.2 MODIS/Aqua Land Surface Temperature (LST)

Surface temperatures near the Amazon River are illustrated using the 8-day 0.05° Aqua MODIS land surface temperature and emissivity product (MYD11C2) available during 2003-2019 (Wan et al., 2015). Retrievals flagged as low quality and cloudy pixels are excluded in the analysis. We assign an equal weight to each month to avoid bias towards dry months. A root mean squared error of 4-5 °C in the Amazon is reported due to deficient cloud detection (Gomis-Cebolla et al., 2018).

2.2.3 CloudSat Cloud Water Content (CWC)

CloudSat 2B-Cloud Water Content-Radar Only (2B-CWC-RO) P1_R05 product (Austin et al., 2009) provides the vertical cloud structure with a 485-m resolution using a 94-GHz cloud profiling radar (Stephens et al., 2002, 2018). Because of its 16-day revisit period and the fixed daytime overpass at 1330 LT in the tropics (Stephens et al., 2002, 2018), the afternoon clouds near the river cannot be fully sampled and deep convection in the late afternoon will be missing. In addition, the liquid cloud water is underestimated during heavy precipitation. With these caveats in mind, we only use typical cases of rainy cloud profiles to illustrate the cloud characteristics near the Amazon River instead of any quantitative analyses.

2.2.4 Weather Research and Forecasting (WRF) model

We performed a regional climate simulation from May 1st, 2007, to July 1st, 2010, using the WRF Advance Research WRF (ARW) version 3.6.1 (Skamarock et al., 2008). The first two months are excluded in the analysis for the model spin-up. The applied parameterization schemes are summarized in Table 1. The single-moment cloud microphysics scheme used here is computationally cheaper than double-moment schemes yet performs reasonably well.

The simulation is conducted with a $10 \text{ km} \times 10 \text{ km}$ domain and 40-s time step. Model outputs are archived every 3 hours. There are 38 vertical layers in the atmosphere up to 10 mb and 4 soil layers. Land cover is identified using the modified IGBP MODIS 21-category data, and the model option of climatological albedo maps is chosen. Initial and boundary conditions are interpolated from the $0.3^{\circ} \times 0.3^{\circ}$ National Centers for Environmental Prediction (NCEP) Climate Forecast System Reanalysis (CFSR) dataset (Saha et al., 2010). Particularly, the river and lake surface temperatures are specified using the previous 3-day average 2-m air temperatures from CFSR.

CFSR outperforms Modern Era Retrospective-Analysis for Research and Applications (MERRA) in simulating the South America water cycle (Quadro et al., 2013), but it overestimates the rainy season precipitation in the Amazon Basin (Blacutt et al., 2015) and underestimates precipitation near the northeastern coast and the Amazon river mouth (Silva et al., 2011). CFSR has also been used to analyze the wind and convergence in the tropical and subtropical South America (Romatschke & Houze, 2013).

Other auxiliary datasets including the land water mask and the surface elevations are introduced in Text S1.

2.3 Methods

2.3.1 Local rainfall percentage anomaly

Considering the northwest-southeast rainfall gradient in this region (Marengo et al., 2001), we defined the local rainfall anomalies as follows to compensate for the large-scale pattern. For each 250-m MOD44W river pixel, the river precipitation is estimated by the average CMORPH rainfall within a 6-km neighborhood, while the baseline precipitation of the surroundings is estimated by the average rainfall of a buffer zone consisting of land pixels 25 - 35 km away from this river pixel. The rainfall percentage anomaly associated with this river pixel is then defined by the percentage difference of the river precipitation relative to the baseline precipitation. This river-forest difference is hereinafter abbreviated as "anomaly", which does not mean the difference from climatology in this paper. We define the land 25 - 35 km away from the rivers as the baseline to avoid too strong impacts of the local river breeze, as is suggested in Dias et al. (2004).

2.3.2 Horizontal moisture convergence

The horizontal moisture convergence at the model output time step just before 1100 LT is computed from the WRF outputs. We choose 1100 LT as the beginning of the daytime period, which is consistent with the diurnal cycle of river rainfall anomalies identified using CMORPH in Section 3. Although convection may initiate earlier than 1100 LT, the previous model output time step around 0800 LT is too early when the daytime river breeze has not yet been established. Our model results also show little difference around 0800 LT between the moisture convergence profiles above the river and the surrounding forest (Figure S2). Similar to the precipitation comparison between the river and the surrounding forest, only the land 25 - 35 km away from the river are used to derive the average moisture convergence and compared against the average moisture convergence over the river.

Moreover, we decompose the horizontal moisture divergence into two terms,

$$\nabla_h \cdot (q\vec{u}) = q\nabla_h \cdot \vec{u} + \vec{u} \cdot \nabla_h q$$

where q is specific humidity [g/kg], and \vec{u} is the horizontal wind vector [m/s]. The subscript h denotes horizontal divergence or horizontal gradient. The first term of this decomposition, or the convergence term, is proportional to horizontal wind divergence, while the second term, or the advection term, is proportional to the gradient of specific humidity. Thus, water vapor diverges when the wind diverges and/or the wind blows from regions of low humidity to regions of high humidity. When considering the Amazon River, the river breeze driven by surface temperature gradient has a component perpendicular to the river (Dias et al., 2004). Therefore, moisture convergence associated with humidity gradient occurs if the specific humidity above the river is lower than the upwind forest. The decomposition is fairly accurate with 3-hourly model outputsWe excluded when the residual of the decomposition is greater than 20% of the total moisture convergence (~XX% of the output).

2.3.3 Humidity change before strong daytime rainfall

Shallow convection is argued to help establish the precondition for deep convection by moistening and destabilizing the free atmosphere in Amazonia (Wright et al., 2017; Zhuang et al., 2017). This "shallow convection moisture pumping" hypothesis is investigated by computing the changes in atmospheric humidity before strong afternoon rain. Consecutive two 3-hourly

model output intervals in the afternoon are selected if (1) there is at least 5-mm precipitation in the later interval, and (2) there is no precipitation in the earlier one. The conditions of the earlier interval are classified as shallow convection (SC), if the maximum cloud fraction between 850 mb and 700 mb is greater than 30%, or no convection (NC) otherwise. Here the cloudiness in the lower troposphere is considered as an indicator of preconditioning shallow convection. A comparison between these SC cases and NC cases help illustrate whether preexisting shallow convection favors deep convection in the afternoon.

3 Satellite observations of rainfall anomalies over the river

3.1 Overview of satellite observed rainfall pattern

Strong surface temperature gradients near the Amazon River are confirmed by MODIS LST products (Figure 1). During the daytime, the annual average LST shows a cool anomaly along most parts of the river (Figure 1a). However, the river has a very small diurnal temperature range, and as a result, it is warmer than the surrounding forest during the nighttime (Figure 1b). This strong surface temperature gradient may induce local circulations that affect convection and rainfall characteristics, as is reported in a few studies (Cohen et al., 2014; Dias et al., 2004). The surface temperature contrast can further be quantified by the Laplacian operator, and warm temperature anomalies are associated with a negative Laplacian (Protter & Weinberger, 2012), vice versa. Except for orographic effects, large positive $\nabla^2 LST$ during the daytime is predominantly found along the river (Figure 1c). During the nighttime, the river also dominates the occurrence of large negative $\nabla^2 LST$ (Figure 1d).

Figure 2a confirms the overall rainfall reduction along the Amazon River according to the 20-year CMORPH data, with different degrees among tributaries. The area with reduced rainfall extends up to several tens of kilometers away from the Negro River and the Solimões River, while the river influence is relatively smaller near the other narrower? tributaries. In addition, the gradient in rainfall appears more symmetric about the Solimões River on both sides but asymmetric about the Negro River, where the north side receives less rain than the south. Compared with similar results presented in Paiva et al. (2011), the spatial pattern of the river influence is better depicted because the CMORPH dataset has a higher resolution.

Daytime precipitation is reduced more significantly over the river (Figure 2b) compared with the total rainfall. Furthermore, the rainfall distribution becomes more asymmetrical on both sides of the Negro River and the Tapajós. Higher rainfall is higher to the east of the Tapajós River while lower rainfall occurs to the west, consistent with results from ground-based S-band radar measurements (Cohen et al. 2014).

In contrast to the strong gradients in daytime rainfall near the river, nighttime rainfall over the river does not substantially differ from the neighboring forests, except for the lower Amazon River near the coast (Figure 2c). The nighttime rainfall increase over the lower Amazon is likely to be associated with squall lines propagating towards southwest overnight (Burleyson et al., 2016; Cohen et al., 1995), and it will be further discussed in Section 3.2. There is a high-

rainfall zone north of the lower Amazon in both Figures 2b and 2c, but its location is further to the west during the nighttime.

The east-west gradient in precipitation and some orographic effects are also noticeable from Figure 2. The western side of the domain is relatively wetter than the eastern side. Remarkably high precipitation is found to the west of the Guiana Highlands (the northwest corner of the domain), which blocks eastward moisture transport and causes a local precipitation minimum to the east.

The rainfall reduction over the river mostly occurs during the daytime and the dry-to-wet transition season (August through October, in Figure 3b). We show the average diurnal and seasonal cycles of the river rainfall anomalies compared with the local surroundings (Figure 3a) and the rainfall percentage anomalies (Figure 3b) across the domain, and find negative values between approximately 1100 and 2300 LT. These negative precipitation anomalies usually occur in more than 70% of all the river grid cells during 1200 - 1800 LT (highlighted by the hatching). The peak percentage reduction (-24.6%) is seen in the afternoon of August – October during the dry-to-wet transition season, and there is a secondary peak in the afternoon of May (-22.6%) associated with the wet-to-dry transition (Figure 3b). By contrast, there are positive rainfall anomalies on average during the nighttime and early morning in December - July, which broadly corresponds to the wet season and the early dry season (Marengo et al., 2001; Zhuang et al., 2017) despite slight differences among tributaries. The daytime precipitation reduction and nighttime increase is consistent with the diurnal pattern of surface temperatures and the river breeze circulation (dos Santos Pinheiro et al., 2014).

3.2 Spatial variability in individual tributaries

The degree to which the rainfall is affected by the river differs among tributaries. This conclusion is further illustrated by comparing the 20-year average rainfall percentage anomalies among the selected tributaries in Figure 4. Overall, rainfall is reduced over most of the rivers, and this reduction is stronger when only daytime rainfall is considered. By contrast, the nighttime anomaly can be either positive or negative depending on the tributary, and the lower Amazon River shows the largest positive anomaly. Except the Amazon, the nighttime percentage anomalies are close to 0 in all the other tributaries.

The diurnal and seasonal cycles of the rainfall anomalies also vary substantially among the tributaries. As relatively wider tributaries, the Negro, the Solimões and the Madeira have similar diurnal and seasonal cycles, with strong reduction in the afternoon and weaker yet mostly negative influences at night (Figures 3c, S3a and S3b).

Unlike the three tributaries mentioned above, the lower Amazon, the Tapajós, the UHE Balbina and the Purus are overall characterized by reduction during the daytime and increase at night (Figures 3d, S3). This type of rainfall anomalies was also reported for the Amazon-Tapajós confluence by Cohen et al. (2014). Strong positive anomalies at night only occur from December to August over the lower Amazon, and there is no substantial difference in nighttime precipitation between the river and the surroundings in other months (Figure 3d). It also shows strongest rainfall reduction during the daytime and strongest increase at night. Over the Tapajós (Figure S3c), rainfall increase can even be found in the afternoon during the dry season (June - August). The hours of rainfall reduction over the UHE Balbina change throughout the year, ranging from about 8 hours in September to almost the entire day in January (Figure S3d). Nocturnal deep convection near the UHE Balbina was also observed by the Amazon Tall Tower Observatory (Oliveira et al., 2020).

We speculate that the nighttime rainfall increase over these tributaries may be related to squall lines which are generated along the sea-breeze front (Burleyson et al., 2016; Cohen et al., 1995; Cutrim et al., 2000; Garstang et al., 1994) and propagate further inland overnight. This is because the spatial and seasonal patterns for coastal squall lines seem to be consistent with the location of these tributaries and the seasonality in Figure 3d. These squall lines have a strong influence on Amazonian rainfall at night during the wet and transitional seasons (Fitzjarrald et al., 2008; Schiro & Neelin, 2019), and the tributaries closer to the Atlantic coast, including the lower Amazon, the Tapajós and the UHE Balbina, have a smaller time delay due to inland propagation (Burleyson et al., 2016). This signal of coastal squall lines is absent in nearly all tributaries further away perhaps because the arrivals of these systems are more or less in phase with daytime heating (Burleyson et al., 2016).

Probably due to its narrow width, it is hard to identify a clear pattern in the diurnal cycle of the Branco River. Nonetheless, we tend to find stronger nighttime rainfall increase from

October to February, and this positive anomaly extends to almost the whole day in February (Figure S3f).

To better illustrate the asymmetric river influence on both sides of the Negro shown in Figure 2c, a typical cloud case on November 6, 2006 is plotted using the CloudSat CWC product (Figure 5). This case shows strong cloud development only to the south of the Negro River. Asymmetric rainfall anomalies were also reported near the Negro-Solimões confluence (Burleyson et al., 2016) and along the Tapajós River (Cohen et al., 2014), together with a single-cell river breeze circulation (Dias et al., 2004). This phenomenon is probably associated with the orientation of the river which is perpendicular to the prevailing winds. It generates convergence on one side of the river and divergence on the other side (Burleyson et al., 2016; Dias et al., 2004; Lu et al., 2005). Dynamic effects due to surface roughness change can favor convection and precipitation on the downwind side of the river, similar to the roughness effects near the small-scale forest-pasture margin (Khanna et al., 2017).

3.3 Suppressed daytime convection due to temperature heterogeneity

We speculate that the daytime rainfall reduction over the rivers is mainly associated with suppressed local convection. This is because a strong reduction in the afternoon is consistent with the diurnal cycle of local surface heat fluxes, and smaller local convective cells are reported to be very common during the daytime in Amazonia (Schiro & Neelin, 2019). MCSs, broadly defined as precipitation cells around or greater than 100 km in at least one dimension (Houze, 2004), contribute substantially to Amazonian precipitation and may also be involved in the daytime rainfall reduction. However, among the major MCS regimes, propagating systems from either the northeastern coast or the northern and eastern part of the Amazon Basin will precipitate at a wide range of local time depending on the location (Cutrim et al., 2000), unlike those fewer locally formed MCS which responds to diurnal heating (Greco et al., 1990). Indeed, detailed examination of the 30-min precipitation maps during the dry-to-wet transition season (Figure S4, using September 8, 2007 as an example) and the wet season (Animation S1, using the first week of April, 2007 as an example) suggests that nighttime rainfall is mostly from MCSs.

Suppressed daytime rainfall due to thermally driven circulation is supported by the close relationship between the Laplacian of surface temperature ($\nabla^2 LST$) and river rainfall anomalies

relative to the surrounding forest. Here the overlapping period 2003-2017 of available precipitation and LST data is chosen to conduct the linear regression between river rainfall anomalies and $\nabla^2 LST$. Instead of absolute temperature, the gradient of temperature directly influences surface wind. Thus, surface convergence is linked to $\nabla^2 LST$ (Duffy et al., 2020; Lindzen & Nigam, 1987).

Except for the UHE Balbina and the lower Amazon, annual average rainfall anomalies of the other tributaries have a significant linear relationship with the climatological monthly average $\nabla^2 LST$ (Figure 6; -0.77±0.13 mm hr⁻¹ °C⁻¹ km²; r = -0.56, p = 2.82e-7, n = 72). A higher $\nabla^2 LST$ is associated with a stronger surface temperature gradient and consequently greater surface divergence over the river. Therefore, the negative slope in Figure 6 confirms more reduction of daytime precipitation where the temperature contrast between the river and the surroundings is stronger. This mechanism is similar to the "Laplacian-of-warming" mechanism in explaining tropical precipitation change under global warming (Duffy et al., 2020), and their results suggest that the relationship could be even stronger when considering the boundary layer virtual temperature rather than surface temperature. Perhaps due to the size or the proximity to the Atlantic coast, the UHE Balbina and the lower Amazon do not follow this negative relationship well (Figure S5).

Despite the negative correlation with the surface temperature pattern, only 31% of the total variance of daytime rainfall reduction can be explained by this relationship. Therefore, other mechanisms also play significant roles in determining the river influence over convection. For example, the surface roughness is drastically different between the water surface and terrestrial vegetation. Although the roughness effects are likely to be minor on a large scale compared with energetics and thermodynamics (Zeng et al., 1996), small-scale heterogeneity of surface roughness between the forest and the pasture has been reported to have a strong dynamic impact over convection in Amazonia (Khanna et al., 2017). Similar impacts are expected for the roughness contrast between the river and the surrounding forest. Moreover, the dynamic effects of the surface roughness can have asymmetrical influence along the prevailing wind direction (Khanna et al., 2017). As is mentioned earlier, some tributaries like the Negro clearly shows convection preference on one side of the river than the other. The location shown in Figure 5 experiences a prevailing northerly wind during the rainy season. Consequently, the lower surface

roughness of the Negro than the forest will favor convection on the downwind side, which is the southern bank. This explanation agrees with the annual daytime precipitation in Figure 2b and the cloud case presented in Figure 5.

4 Model simulation of rainfall anomalies over the river

374

375

376

377

378

379

380

381

382

383

384

385

386

387

388

389

390

391

392

393

394

395

396

397

398

399

400

401

402

4.1 Overview of simulated rainfall pattern

The overall surface temperature patterns simulated by WRF look similar to MODIS retrievals, but the absolute values are around 5°C too warm during the day and around 3°C too cold at night (Figure S6). The actual descrepancy may be small because modeled daytime or nighttime temperature is the average skin temperature over 12 hours centered at the satellite overpass (0200 and 1400 LT), while MODIS daytime or nighttime temperature is a composite average of Aqua swaths. MODIS surface temperatures in Amazonia also suffers the issue due to heavy cloudiness (Gomis-Cebolla et al., 2018). Additionally, the prescribed river temperature in WRF is cooler than MODIS during the day and warmer than MODIS during the night. As a result, the average surface temperature anomalies of the WRF river grid boxes relative to neighboring forest grid boxes 10 km away are exaggerated to -8.4 °C during the day and +6.6 °C at night. High values of modeled ∇^2 LST are also concentrated near the Amazon River (Figures 7c, d). The river grid cells are discontinuous or even completely missing for those tributaries narrower than the 10-km horizontal grid resolution, such as the Solimões, the Madeira, the Branco and the Purus. Therefore, the following analyses will mainly focus on those wider tributaries resolved by the model grid, such as the Negro, the Tapajós and the lower Amazon. Despite the exaggerated surface temperature gradient near the river in the model results, the influence of surface temperature on convection and precipitation is expected to be qualitatively similar compared with observations.

The spatial distribution of surface energy fluxes corresponds with surface temperatures (Figure S7). During the daytime at 1400 LT, the cooler river has a sensible heat flux an order of magnitude lower than the rainforest. This difference in sensible heat flux is so strong that even some narrow tributaries, such as the upper part of the Tapajós, can be visually seen from Figure S7a. Although the daytime latent heat flux of the river is not as small as the sensible heat flux, it is still substantially lower than the surrounding forest. The river-forest contrast in surface energy

fluxes is reversed at night, when the river has both higher sensible and latent heat fluxes. The forest has nearly zero or slightly negative sensible and latent heat fluxes at 0200 LT. The nighttime sensible heat flux of the river is also small, but the latent heat flux can reach around 100 W/m² in some tributaries. A higher surface flux helps destabilize the atmosphere and facilitates local deep convection (Wu & Lee, 2019; Zhuang et al., 2017), and it also has the potential to drive changes in the mesoscale and large-scale transport (Wright et al., 2017). In addition to the absolute amount of surface energy fluxes, the Bowen ratio of sensible heat flux to latent heat flux is higher over the forest than the river. A higher daytime Bowen ratio is also found to favor deeper convective clouds in the Amazon region (Wu & Lee, 2019).

The modeled large-scale rainfall pattern resembles the observations, except the unrealistically high nighttime rainfall in the southwest corner of the domain (Figures 8c and 8d). Daytime rainfall is substantially suppressed over the river in our simulation (Figure 8c), and the majority of this precipitation anomaly is attributed to convective rainfall rather than large-scale rainfall (Figure 8e). However, the absolute values of rainfall are about twice as high as the CMORPH data across the domain, and nighttime rainfall in the southern part of the domain is further overestimated compared with observations (Figure 8d).

Compared with the observations in Figure 2c, a zone of relatively higher nighttime rainfall extending northwards from the lower Amazon is successfully captured in our simulation (Figures 8c and 8d). As is discussed in previous sections, this zone is associated with the observed positive nighttime rainfall anomalies over the lower Amazon during December to August, probably as a result of propagating squall lines. The nighttime propagation of squall lines is also simulated in the model (Figure S8). This zone of higher nighttime rainfall covering part of the lower Amazon is also a seasonal phenomenon in WRF results, which starts to appear in October and migrates northwards away from the river in April. A westward propagation of this high-rainfall zone from day to night can also be seen in Figure 8, but not as clear as in Figure 2. By contrast, the daytime rainfall suppression over the Amazon River in our simulation is noticeable every month (Figure S9), similar to CMORPH measurements.

4.2 Horizontal moisture convergence before daytime rainfall

In addition to differences in rainfall amount, the river and its surrounding forest are characterized by distinct moisture convergence profiles before daytime rain events (Figure 9).

Here all the daytime rain events are categorized based on the total rainfall over the 12-hour period from 1100 to 2300 LT, and the convergence is computed for the closest model output time step before 1100 LT. For weaker to moderate rain (< 20 mm / 12 hr) over the river, the model simulation indicates moisture divergence near the surface and moisture convergence above up to approximately 750 mb (Figure 9a). With increasing precipitation, changes occur below 850 mb where the divergence near the surface weakens and finally disappears and the convergence strengthens. When the rainfall exceeds 20 mm / 12 hr, moisture convergence extends from the surface up to 700 mb or even higher. Furthermore, together with higher and higher rainfall, the increase in moisture convergence extends above 850 mb.

In contrast to the separate convergence/divergence layers above the river, moisture convergence, if existing, is always highest near the surface over the surrounding forest (Figure 9b). When the rainfall exeeds 5 mm / 12 hr, the moisture convergence layer extends from the surface into the lower free troposphere, and this layer deepens with precipitation intensity. Similar to the river, the increase in moisture convergence first takes place below 850 mb, and the rises to higher levels. The deepening surface convergence layer may also relate to the rapid pickup of the well-known nonlinear precipitation-moisture curve (Bretherton et al., 2004; Neelin et al., 2009; Peters & Neelin, 2006; Schiro & Neelin, 2019), which may be worth further investigation in different environmental conditions.

The moisture convergence profiles for daytime rain events are then decomposed using the equation described in Section 2.3.2 (Figure 10). The moisture advection component $(\vec{u} \cdot \nabla_h q)$ is insensitive to the amount of precipitation, and most changes in Figure 9 are contributed by the mass convergence component $(q\nabla_h \cdot \vec{u})$. Although the mass convergence term mainly controls the shape of the total moisture convergence profile, moisture convergence due to the specific humidity gradient and moisture advection is also important for the river (Figure 10a). The separate convergence and divergence layers are results of the separate layers of mass convergence above the surface and divergence near the surface. The total moisture convergence profile is further modulated by the moisture advection component. Because the specific humidity is lower above the river than the surrounding forest, the advection component is always converging below 850 mb and reaches its peak near the boundary layer top, where is contributes to nearly half of the total moisture convergence. The shape of the wind convergence profile is

associated with the river breeze circulation, the surface temperature gradient and subsidence in the lower troposphere over the cool river surface (Dias et al., 2004).

Over the surrounding forest, the total moisture convergence profile follows wind convergence and the advection component related to the specific humidity gradient is much smaller (Figure 10b). The average boundary layer top is higher above the forest (880 mb), and both moisture and wind convergence occur mostly within the boundary layer. There is weak and insignificant moisture or wind divergence above the boundary layer, indicating only a minor influence of the free-tropospheric moisture supply (Wu & Lee, 2019). The river breeze circulation during the daytime may strengthen this type of profile near the Amazon River. The advection component contributes to a significant but small moisture divergence below 800 mb.

4.3 Role of shallow convection

463

464

465

466

467

468

469

470

471

472

473

474

475

476

477

478

479

480

481

482

483

484

485

486

487

488

489

490

491

The "shallow convection moisture pumping" mechanism is not explicitly supported in our model results, where shallow convection ahead of strong afternoon rainfall does not lead to more atmospheric moistening locally (Figure 11). Both the river and the surrounding forest show similar comparisons between the humidity changes in SC cases and in NC cases. The vertical profiles of specific humidity changes are characterized by an increase near the surface and a decrease above. However, for NC cases, the increasing specific humidity can be found util 800 mb, and the humidity decrease above is relatively small. The peak increase occurs around 900 mb, and is higher above the river than above the forest. By contrast, the humidity increase when there is shallow convection is limited below 900 mb and is small. The moisture divergence above reaches its peak near 800 mb and extends upwards until 500 mb. Particularly for the river, the preexisting shallow convection does not contribute to any significantly higher values across the troposphere. Despite a much lower humidity increase near the surface, the shallow convection above the forest reduce the humidity decrease above 700 mb, which may be favorable for deep convection to some extent when considering the lateral entrainment along the convecting path. To test the sensitivity of our results, we have also tried 10-mm precipitation as the threshold to select strong rainfall cases, and the result is very similar (Figure S10). Nevertheless, our results cannot exclude the possibility that shallow convection in earlier days or at locations nearby could contribute to a moistened atmosphere.

5 Synthesis and conclusions

Precipitating convection over the Amazon River and the surrounding forest is compared in this study. The influence of synoptic conditions is minimized in this comparison because of the short distance between the river and the forest, and hence the effects due to surface conditions can be isolated. Our results particularly highlight the role of surface temperatures, which suppress daytime convection over the river, interact with nighttime convection near the coast, and cause the double-layer moisture convergence/divergence structure.

Satellite-based analyses show rainfall reduction during the daytime across the domain and throughout the year, and substantial reduction only occurs in the afternoon in most tributaries when surface-induced convection is expected to reach its peak (Zipser et al., 2006). This reduction is particularly important during the dry-to-wet transition season, when large-scale moisture supply has not yet started to increase (Fu & Li, 2004; Wright et al., 2017) and the average convective intensity is highest in a year (D. Wang et al., 2018; Wu & Lee, 2019). The uniqueness of the dry-to-wet transition season could come from a combination of increasing humidity and yet elevated instability during the daytime (Wu & Lee, 2019; Zhuang et al., 2017), as well as enhanced wind shear (Giangrande et al., 2020). Terrestrial evapotranspiration is reported to be responsible for the early increase in atmospheric moisture (Wright et al., 2017), and the developing deep convection further triggers large-scale moisture advection from the ocean (Fu et al., 1999).

The magnitude of daytime reduction has a significant linear relationship with the Laplacian of surface temperature. All these characteristics support the daytime suppression of local convection above the river due to thermally driven divergence (Duffy et al., 2020; Lindzen & Nigam, 1987). The "Laplacian-of-warming" mechanism is reported to be important in oceanic precipitation responses to climate change (Duffy et al., 2020), and this study also confirms the close connection between $\nabla^2 LST$ and precipitation near the Amazon River on land.

The river also increases rainfall at night but only in a few tributaries, such as the Amazon-Tapajós confluence (Cohen et al., 2014). Strongest nighttime signal is found in the lower Amazon River near the Atlantic coast, which forms the southern end of a high-rainfall corridor associated with propagating squall lines (Burleyson et al., 2016; Cohen et al., 1995).

521

522

523

524

525

526

527

528

529

530

531

532

533

534

535

536

537

538

539

540

541

542

543

544

545

546

The WRF model further show that moisture convergence before daytime rain above the surface is predominantly contributed by wind convergence as well as the humidity gradient near the boundary top. By contrast, the surrounding forest not far away gathers moisture primarily through convergence within the boundary layer or a little above, as is known from previous studies (Schiro & Neelin, 2019; Wu & Lee, 2019). In addition to the knowledge of surface-convection interaction, this study also has implications on the Observations and Modeling of the Green Ocean Amazon (GoAmazon2014/5) Experiment (Martin et al., 2016). The T3 site of GoAmazon2014/5 is reported to be slightly suppressed by the river breeze of the Negro (Burleyson et al., 2016). Our results regarding the unique convective characteristics near the Amazon River may further help the interpretation of the GoAmazon datasets. **Acknowledgments and Data** Data is available through Amatulli et al. (2018), Austin et al. (2009) at http://www.cloudsat.cira.colostate.edu/data-products/level-2b/2b-cwc-ro/ (version P1 R05), Carroll et al. (2017), Joyce et al. (2004) at http://www.ftpstatus.com/dir properties.php?sname=ftp.cpc.ncep.noaa.gov&did=8/, Mayorga et al. (2012), Saha et al. (2010), Wan et al. (2015) and Wu et al. (2020). This work was supported by NSF Climate & Large-Scale Dynamics (NSF AGS 1944545). This paper has been authored by employees of Brookhaven Science Associates, LLC, under contract DE-SC0012704 with the U.S. DOE. The publisher by accepting the paper for publication acknowledges that the U.S. Government retains a nonexclusive, paid-up, irrevocable, worldwide license to publish or reproduce the published form of this paper, or allow others to do so, for U.S. Government purposes. We want to thank P. Hall, H. Kershaw and the Center for Computation and Visualization, Brown University for their computational support. We also want to thank the reviewers for their suggestions.

547 References

- 548 Amatulli, G., Domisch, S., Tuanmu, M.-N., Parmentier, B., Ranipeta, A., Malczyk, J., & Jetz, W.
- 549 (2018). A suite of global, cross-scale topographic variables for environmental and biodiversity
- 550 modeling. Scientific data, 5, 180040. https://doi.org/10.1038/sdata.2018.40
- Andreae, M. O., Rosenfeld, D., Artaxo, P., Costa, A. A., Frank, G. P., Longo, K. M., & Silva-
- 552 Dias, M. A. F. (2004). Smoking rain clouds over the Amazon. *Science*, 303(5662), 1337-1342.
- 553 https://doi.org/10.1126/science.1092779
- Austin, R. T., Heymsfield, A. J., & Stephens, G. L. (2009). Retrieval of ice cloud microphysical
- parameters using the CloudSat millimeter-wave radar and temperature. *Journal of Geophysical*
- 556 Research-Atmospheres, 114(D8). https://doi.org/10.1029/2008jd010049
- Bagley, J. E., Desai, A. R., Harding, K. J., Snyder, P. K., & Foley, J. A. (2014). Drought and
- deforestation: Has land cover change influenced recent precipitation extremes in the Amazon?
- 559 Journal of Climate, 27(1), 345-361. https://doi.org/10.1175/Jcli-D-12-00369.1
- Bang, S. D., & Zipser, E. J. (2016). Seeking reasons for the differences in size spectra of
- electrified storms over land and ocean. Journal of Geophysical Research-Atmospheres, 121(15),
- 562 9048-9068. https://doi.org/10.1002/2016jd025150
- Blacutt, L. A., Herdies, D. L., de Gonçalves, L. G. G., Vila, D. A., & Andrade, M. (2015).
- Precipitation comparison for the CFSR, MERRA, TRMM3B42 and Combined Scheme datasets
- 565 in Bolivia. Atmospheric Research, 163, 117-131. https://doi.org/10.1016/j.atmosres.2015.02.002
- Bretherton, C. S., Peters, M. E., & Back, L. E. (2004). Relationships between water vapor path
- and precipitation over the tropical oceans. *Journal of Climate*, 17(7), 1517-1528.
- 568 https://doi.org/10.1175/1520-0442(2004)017<1517:Rbwvpa>2.0.Co;2
- 569 Burleyson, C. D., Feng, Z., Hagos, S. M., Fast, J., Machado, L. A. T., & Martin, S. T. (2016).
- 570 Spatial Variability of the Background Diurnal Cycle of Deep Convection around the
- GoAmazon2014/5 Field Campaign Sites. *Journal of Applied Meteorology and Climatology*,
- 572 55(7), 1579-1598. https://doi.org/10.1175/Jamc-D-15-0229.1
- 573 Carroll, M., DiMiceli, C. M., Wooten, M. R., Hubbard, A. B., Sohlberg, R. A., & Townshend, J.
- R. G. (2017). MOD44W: MODIS/Terra Land Water Mask Derived from MODIS and SRTM L3
- Global 250m SIN Grid V006. NASA Earth Observing System Data and Information System Land

- 576 Process Distributed Active Archive Centers, Sioux Falls, SD.
- 577 https://doi.org/10.5067/MODIS/MOD44W.006
- 578 Chen, F., & Dudhia, J. (2001). Coupling an advanced land surface-hydrology model with the
- Penn State-NCAR MM5 modeling system. Part I: Model implementation and sensitivity.
- 580 *Monthly Weather Review*, 129(4), 569-585. https://doi.org/10.1175/1520-
- 581 0493(2001)129<0569:Caalsh>2.0.Co;2
- Cohen, J. C., Silva Dias, M. A., & Nobre, C. A. (1995). Environmental conditions associated
- with Amazonian squall lines: A case study. *Monthly Weather Review*, 123(11), 3163-3174.
- 584 https://doi.org/10.1175/1520-0493(1995)123<3163:ECAWAS>2.0.CO;2
- Cohen, J. C. P., Fitzjarrald, D. R., D'Oliveira, F. A. F., Saraiva, I., Barbosa, I. R. d. S., Gandu, A.
- 586 W., & Kuhn, P. A. (2014). Radar-observed spatial and temporal rainfall variability near the
- Tapajós-Amazon confluence. Revista Brasileira de Meteorologia, 29(SPE), 23-30.
- 588 https://doi.org/10.1590/0102-778620130058
- Collins, W. D., Rasch, P. J., Boville, B. A., Hack, J. J., McCaa, J. R., Williamson, D. L., et al.
- 590 (2004). Description of the NCAR community atmosphere model (CAM 3.0). NCAR Tech. Note
- 591 *NCAR/TN-464+ STR*, 226.
- Cutrim, E. M., Martin, D. W., Butzow, D. G., Silva, I. M., & Yulaeva, E. (2000). Pilot analysis
- of hourly rainfall in central and eastern Amazonia. *Journal of Climate*, 13(7), 1326-1334.
- 594 https://doi.org/10.1175/1520-0442(2000)013<1326:PAOHRI>2.0.CO;2
- 595 Dias, M. A. F. S., Dias, P. L. S., Longo, M., Fitzjarrald, D. R., & Denning, A. S. (2004). River
- 596 breeze circulation in eastern Amazonia: observations and modelling results. *Theoretical and*
- 597 Applied Climatology, 78(1-3), 111-121. https://doi.org/10.1007/s00704-004-0047-6
- dos Santos Pinheiro, G. M., Poitrasson, F., Sondag, F., Cochonneau, G., & Vieira, L. C. (2014).
- 599 Contrasting iron isotopic compositions in river suspended particulate matter: the Negro and the
- Amazon annual river cycles. Earth and Planetary Science Letters, 394, 168-178.
- 601 https://doi.org/10.1016/j.epsl.2014.03.006
- Duffy, M. L., O'Gorman, P. A., & Back, L. E. (2020). Importance of Laplacian of low-level
- warming for the response of precipitation to climate change over tropical oceans. *Journal of*
- 604 *Climate*, 33(10), 4403-4417. https://doi.org/10.1175/JCLI-D-19-0365.1

- Fitzjarrald, D. R., Sakai, R. K., Moraes, O. L. L., de Oliveira, R. C., Acevedo, O. C., Czikowsky,
- 606 M. J., & Beldini, T. (2008). Spatial and temporal rainfall variability near the Amazon-Tapajos
- 607 confluence. *Journal of Geophysical Research-Biogeosciences*, 113(G1).
- 608 https://doi.org/10.1029/2007jg000596
- Fu, R., & Li, W. (2004). The influence of the land surface on the transition from dry to wet
- season in Amazonia. *Theoretical and Applied Climatology*, 78(1-3), 97-110.
- 611 https://doi.org/10.1007/s00704-004-0046-7
- Fu, R., Zhu, B., & Dickinson, R. E. (1999). How do atmosphere and land surface influence
- seasonal changes of convection in the tropical amazon? *Journal of Climate*, 12(5), 1306-1321.
- 614 doi:10.1175/1520-0442(1999)012<1306:Hdaals>2.0.Co;2
- Garstang, M., Massie Jr, H. L., Halverson, J., Greco, S., & Scala, J. (1994). Amazon coastal
- 616 squall lines. Part I: Structure and kinematics. *Monthly Weather Review*, 122(4), 608-622.
- 617 https://doi.org/10.1175/1520-0493(1994)122<0608:ACSLPI>2.0.CO;2
- 618 Gentine, P., Holtslag, A. A. M., D'Andrea, F., & Ek, M. (2013). Surface and Atmospheric
- 619 Controls on the Onset of Moist Convection over Land. Journal of Hydrometeorology, 14(5),
- 620 1443-1462. https://doi.org/10.1175/Jhm-D-12-0137.1
- Gentine, P., Massmann, A., Lintner, B. R., Alemohammad, S. H., Fu, R., Green, J. K., et al.
- 622 (2019). Land-atmosphere interactions in the tropics a review. *Hydrology and Earth System*
- 623 Sciences, 23(10), 4171-4197. https://doi.org/10.5194/hess-23-4171-2019
- 624 Giangrande, S. E., Wang, D., & Mechem, D. B. (2020). Cloud regimes over the Amazon Basin:
- perspectives from the GoAmazon2014/5 campaign. Atmospheric Chemistry and Physics, 20(12),
- 626 7489-7507. https://doi.org/10.5194/acp-20-7489-2020
- 627 Gomis-Cebolla, J., Jimenez, J. C., & Sobrino, J. A. (2018). LST retrieval algorithm adapted to
- the Amazon evergreen forests using MODIS data. Remote Sensing of Environment, 204, 401-
- 629 411. https://doi.org/10.1016/j.rse.2017.10.015
- 630 Greco, S., Scala, J., Halverson, J., Massie Jr, H. L., Tao, W.-K., & Garstang, M. (1994). Amazon
- 631 coastal squall lines. Part II: Heat and moisture transports. *Monthly Weather Review*, 122(4), 623-
- 632 635. https://doi.org/10.1175/1520-0493(1994)122<0623:ACSLPI>2.0.CO;2

- 633 Greco, S., Swap, R., Garstang, M., Ulanski, S., Shipham, M., Harriss, R. C., et al. (1990).
- Rainfall and surface kinematic conditions over central Amazonia during ABLE 2B. Journal of
- 635 Geophysical Research: Atmospheres, 95(D10), 17001-17014.
- 636 https://doi.org/10.1029/JD095iD10p17001
- Hohenegger, C., & Stevens, B. (2018). The role of the permanent wilting point in controlling the
- 638 spatial distribution of precipitation. Proceedings of the National Academy of Sciences of the
- 639 *United States of America*, 115(22), 5692-5697. https://doi.org/10.1073/pnas.1718842115
- Holloway, C. E., & Neelin, J. D. (2009). Moisture Vertical Structure, Column Water Vapor, and
- Tropical Deep Convection. *Journal of the Atmospheric Sciences*, 66(6), 1665-1683.
- 642 https://doi.org/10.1175/2008jas2806.1
- Hong, S. Y., Dudhia, J., & Chen, S. H. (2004). A revised approach to ice microphysical
- processes for the bulk parameterization of clouds and precipitation. *Monthly Weather Review*,
- 645 132(1), 103-120. https://doi.org/10.1175/1520-0493(2004)132<0103:Aratim>2.0.Co;2
- Hong, S. Y., Noh, Y., & Dudhia, J. (2006). A new vertical diffusion package with an explicit
- treatment of entrainment processes. *Monthly Weather Review*, 134(9), 2318-2341.
- 648 https://doi.org/10.1175/Mwr3199.1
- Houze, R. A. (2004). Mesoscale convective systems. Reviews of Geophysics, 42(4).
- 650 https://doi.org/10.1029/2004rg000150
- 651 Itterly, K. F., Taylor, P. C., & Dodson, J. B. (2018). Sensitivity of the Amazonian convective
- diurnal cycle to its environment in observations and reanalysis. *Journal of Geophysical*
- 653 Research: Atmospheres, 123(22), 12-621. https://doi.org/10.1029/2018JD029251
- Joyce, R. J., Janowiak, J. E., Arkin, P. A., & Xie, P. P. (2004). CMORPH: A method that
- produces global precipitation estimates from passive microwave and infrared data at high spatial
- and temporal resolution. *Journal of Hydrometeorology*, 5(3), 487-503.
- 657 https://doi.org/10.1175/1525-7541(2004)005<0487:Camtpg>2.0.Co;2
- Kain, J. S. (2004). The Kain-Fritsch convective parameterization: An update. *Journal of Applied*
- 659 *Meteorology*, 43(1), 170-181. https://doi.org/10.1175/1520-
- 660 0450(2004)043<0170:Tkcpau>2.0.Co;2

- Khanna, J., Medvigy, D., Fueglistaler, S., & Walko, R. (2017). Regional dry-season climate
- changes due to three decades of Amazonian deforestation. Nature Climate Change, 7(3), 200-
- 663 204. https://doi.org/10.1038/Nclimate3226
- Kuo, Y. H., Neelin, J. D., & Mechoso, C. R. (2017). Tropical Convective Transition Statistics
- and Causality in the Water Vapor-Precipitation Relation. *Journal of the Atmospheric Sciences*,
- 666 74(3), 915-931. https://doi.org/10.1175/Jas-D-16-0182.1
- Lee, J.-E., Oliveira, R. S., Dawson, T. E., & Fung, I. (2005). Root functioning modifies seasonal
- 668 climate. Proceedings of the National Academy of Sciences of the United States of America,
- 669 102(49), 17576-17581. https://doi.org/10.1073/pnas.0508785102
- Lindzen, R. S., & Nigam, S. (1987). On the role of sea surface temperature gradients in forcing
- low-level winds and convergence in the tropics. *Journal of the Atmospheric Sciences*, 44(17),
- 672 2418-2436. https://doi.org/10.1175/1520-0469(1987)044<2418:OTROSS>2.0.CO;2
- Lu, L. X., Denning, A. S., da Silva-Dias, M. A., da Silva-Dias, P., Longo, M., Freitas, S. R., &
- Saatchi, S. (2005). Mesoscale circulations and atmospheric CO2 variations in the Tapajos
- Region, Para, Brazil. Journal of Geophysical Research-Atmospheres, 110(D21).
- 676 https://doi.org/10.1029/2004jd005757
- Lucas, C., Zipser, E. J., & Lemone, M. A. (1994). Convective available potential energy in the
- environment of oceanic and continental clouds Correction and comments. *Journal of the*
- 679 Atmospheric Sciences, 51(24), 3829-3830. https://doi.org/10.1175/1520-
- 680 0469(1994)051<3829:Capeit>2.0.Co;2
- Marengo, J. A., Liebmann, B., Kousky, V. E., Filizola, N. P., & Wainer, I. C. (2001). Onset and
- end of the rainy season in the Brazilian Amazon Basin. *Journal of Climate*, 14(5), 833-852.
- 683 https://doi.org/10.1175/1520-0442(2001)014<0833:Oaeotr>2.0.Co;2
- Martin, S. T., Artaxo, P., Machado, L. A. T., Manzi, A. O., Souza, R. A. F., Schumacher, C., et
- al. (2016). Introduction: Observations and Modeling of the Green Ocean Amazon
- 686 (GoAmazon2014/5). *Atmospheric Chemistry and Physics*, 16(8), 4785-4797.
- 687 https://doi.org/10.5194/acp-16-4785-2016

- 688 Mayorga, E., Logsdon, M. G., Ballester, M. V. R., & Richey, J. E. (2012). LBA-ECO CD-06
- Amazon River basin land and stream drainage direction maps. ORNL DAAC.
- 690 https://doi.org/10.3334/ORNLDAAC/1086
- Nepstad, D. C., de Carvalho, C. R., Davidson, E. A., Jipp, P. H., Lefebvre, P. A., Negreiros, G.
- H., et al. (1994). The role of deep roots in the hydrological and carbon cycles of Amazonian
- 693 forests and pastures. *Nature*, 372, 666-669. https://doi.org/10.1038/372666a0
- Neelin, J. D., Peters, O., & Hales, K. (2009). The Transition to Strong Convection. Journal of the
- 695 Atmospheric Sciences, 66(8), 2367-2384. https://doi.org/10.1175/2009jas2962.1
- Oliveira, M. I., Acevedo, O. C., Sörgel, M., Nascimento, E. L., Manzi, A. O., Oliveira, P. E., et
- al. (2020). Planetary boundary layer evolution over the Amazon rainforest in episodes of deep
- 698 moist convection at the Amazon Tall Tower Observatory. Atmospheric Chemistry and Physics,
- 699 20(1), 15-27. https://doi.org/10.5194/acp-2019-373
- Paiva, R. C. D., Buarque, D. C., Clarke, R. T., Collischonn, W., & Allasia, D. G. (2011).
- Reduced precipitation over large water bodies in the Brazilian Amazon shown from TRMM data.
- 702 *Geophysical Research Letters*, 38(4). https://doi.org/10.1029/2010gl045277
- Peters, O., & Neelin, J. D. (2006). Critical phenomena in atmospheric precipitation. *Nature*
- 704 *Physics*, 2(6), 393-396. https://doi.org/10.1038/nphys314
- Protter, M. H., & Weinberger, H. F. (2012). *Maximum principles in differential equations*.
- 706 Springer Science & Business Media.
- 707 Quadro, M. F., Berbery, E. H., Silva Dias, M. A., Herdies, D. L., & Gonçalves, L. G. (2013,
- May). The atmospheric water cycle over South America as seen in the new generation of global
- 709 reanalyses. AIP Conference Proceedings, 1531(1), 732-735. https://doi.org/10.1063/1.4804874
- Romatschke, U., & Houze Jr, R. A. (2013). Characteristics of precipitating convective systems
- accounting for the summer rainfall of tropical and subtropical South America. *Journal of*
- 712 *Hydrometeorology*, 14(1), 25-46. https://doi.org/10.1175/JHM-D-12-060.1
- Running, S., Mu, Q., Zhao, M. (2017). MYD16A2 MODIS/Aqua Net Evapotranspiration 8-Day
- 714 L4 Global 500m SIN Grid V006 [Data set]. NASA EOSDIS Land Processes DAAC.
- 715 https://doi.org/10.5067/MODIS/MYD16A2.006

- Saha, S., et al. 2010. NCEP Climate Forecast System Reanalysis (CFSR) Selected Hourly Time-
- Series Products, January 1979 to December 2010. Research Data Archive at the National Center
- 718 for Atmospheric Research, Computational and Information Systems Laboratory.
- 719 https://doi.org/10.5065/D6513W89
- 720 Schiro, K. A., Ahmed, F., Giangrande, S. E., & Neelin, J. D. (2018). GoAmazon2014/5
- campaign points to deep-inflow approach to deep convection across scales. *Proceedings of the*
- National Academy of Sciences of the United States of America, 115(18), 4577-4582.
- 723 https://doi.org/10.1073/pnas.1719842115
- Schiro, K. A., & Neelin, J. D. (2019). Deep Convective Organization, Moisture Vertical
- Structure, and Convective Transition Using Deep-Inflow Mixing. *Journal of the Atmospheric*
- 726 Sciences, 76(4), 965-987. https://doi.org/10.1175/Jas-D-18-0122.1
- 727 Silva, V. B., Kousky, V. E., & Higgins, R. W. (2011). Daily precipitation statistics for South
- America: An intercomparison between NCEP reanalyses and observations. *Journal of*
- 729 *Hydrometeorology*, 12(1), 101-117. https://doi.org/10.1175/2010JHM1303.1
- Skamarock, W. C., Klemp, J. B., Dudhia, J., Gill, D. O., Barker D. M., Duda, M. G., et al.
- 731 (2008). A description of the advanced research WRF version 3. National Center for Atmospheric
- 732 Research, Boulder, CO.
- 733 Stephens, G., Winker, D., Pelon, J., Trepte, C., Vane, D., Yuhas, C., et al. (2018). CloudSat and
- 734 CALIPSO within the A-Train: Ten Years of Actively Observing the Earth System. *Bulletin of*
- 735 the American Meteorological Society, 99(3), 569-581. https://doi.org/10.1175/Bams-D-16-
- 736 0324.1
- 737 Stephens, G. L., Vane, D. G., Boain, R. J., Mace, G. G., Sassen, K., Wang, Z. E., et al. (2002).
- 738 The CloudSat mission and the A-Train A new dimension of space-based observations of clouds
- and precipitation. Bulletin of the American Meteorological Society, 83(12), 1771-1790.
- 740 https://doi.org/10.1175/Bams-83-12-1771
- Takahashi, H., Luo, Z. J., & Stephens, G. L. (2017). Level of neutral buoyancy, deep convective
- outflow, and convective core: New perspectives based on 5 years of CloudSat data. *Journal of*
- 743 *Geophysical Research: Atmospheres*, 122(5), 2958-2969. https://doi.org/10.1002/2016JD025969

- Taylor, C. M., de Jeu, R. A. M., Guichard, F., Harris, P. P., & Dorigo, W. A. (2012). Afternoon
- 745 rain more likely over drier soils. *Nature*, 489(7416), 423-426.
- 746 https://doi.org/10.1038/nature11377
- 747 Taylor, C. M., & Ellis, R. J. (2006). Satellite detection of soil moisture impacts on convection at
- the mesoscale. *Geophysical Research Letters*, 33(3). https://doi.org/10.1029/2005gl025252
- 749 Tian, Y., & Peters-Lidard, C. D. (2007). Systematic anomalies over inland water bodies in
- satellite-based precipitation estimates. Geophysical Research Letters, 34(14).
- 751 https://doi.org/10.1029/2007GL030787
- Wall, C., Liu, C. T., & Zipser, E. (2013). A climatology of tropical congestus using CloudSat.
- *Journal of Geophysical Research-Atmospheres*, 118(12), 6478-6492.
- 754 https://doi.org/10.1002/jgrd.50455
- Wan, Z., Hook, S., & Hulley, G. (2015). MYD11C2 MODIS/Aqua Land Surface
- 756 Temperature/Emissivity 8-Day L3 Global 0.05Deg CMG V006 [Data set]. NASA EOSDIS Land
- 757 Processes DAAC. https://doi.org/10.5067/MODIS/MYD11C2.006
- Wang, D., Giangrande, S. E., Bartholomew, M. J., Hardin, J., Feng, Z., Thalman, R., &
- Machado, L. A. T. (2018). The Green Ocean: precipitation insights from the GoAmazon2014/5
- 760 experiment. Atmospheric Chemistry and Physics, 18(12), 9121-9145.
- 761 https://doi.org/10.5194/acp-18-9121-2018
- Wang, D., Giangrande, S. E., Feng, Z., Hardin, J. C., & Prein, A. F. (2020). Updraft and
- downdraft core size and intensity as revealed by radar wind profilers: MCS observations and
- 764 idealized model comparisons. Journal of Geophysical Research: Atmospheres, 125,
- 765 e2019JD031774. https://doi.org/10.1029/2019JD031774
- Wang, D., Giangrande, S. E., Schiro, K., Jensen, M. P., & Houze, R. A. (2019). The
- characteristics of tropical and midlatitude mesoscale convective systems as revealed by radar
- wind profilers. *Journal of Geophysical Research: Atmospheres*, 124, 4601-4619.
- 769 https://doi.org/10.1029/2018JD030087
- Wang, J. F., Chagnon, F. J. F., Williams, E. R., Betts, A. K., Renno, N. O., Machado, L. A. T., et
- al. (2009). Impact of deforestation in the Amazon basin on cloud climatology. *Proceedings of the*

- National Academy of Sciences of the United States of America, 106(10), 3670-3674.
- 773 https://doi.org/10.1073/pnas.0810156106
- Williams, E., Rosenfeld, D., Madden, N., Gerlach, J., Gears, N., Atkinson, L., et al. (2002).
- 775 Contrasting convective regimes over the Amazon: Implications for cloud electrification. *Journal*
- of Geophysical Research-Atmospheres, 107(D20). https://doi.org/10.1029/2001jd000380
- Wright, J. S., Fu, R., Worden, J. R., Chakraborty, S., Clinton, N. E., Risi, C., et al. (2017).
- Rainforest-initiated wet season onset over the southern Amazon. *Proceedings of the National*
- 779 Academy of Sciences of the United States of America, 114(32), 8481-8486.
- 780 https://doi.org/10.1073/pnas.1621516114
- Wu, M., & Lee, J.-E. (2019). Thresholds for Atmospheric Convection in Amazonian Rainforests.
- 782 *Geophysical Research Letters*, 46(16), 10024-10033. https://doi.org/10.1029/2019gl082909
- Wu, M., Lee, J.-E., Wang, D., & Salameh, M. (2020). Data from "Suppressed Daytime
- Convection over the Amazon River". Brown Digital Repository. https://doi.org/10.26300/y2m9-
- 785 mk21
- Zeng, N., Dickinson, R. E., & Zeng, X. (1996). Climatic impact of Amazon deforestation—A
- mechanistic model study. Journal of Climate, 9(4), 859-883. https://doi.org/10.1175/1520-
- 788 0442(1996)009<0859:CIOADM>2.0.CO;2
- Zhang, Y., & Klein, S. A. (2010). Mechanisms affecting the transition from shallow to deep
- 790 convection over land: Inferences from observations of the diurnal cycle collected at the ARM
- 791 Southern Great Plains site. *Journal of the Atmospheric Sciences*, 67(9), 2943-2959.
- 792 https://doi.org/10.1175/2010JAS3366.1
- 793 Zhuang, Y., Fu, R., Marengo, J. A., & Wang, H. (2017). Seasonal variation of shallow-to-deep
- 794 convection transition and its link to the environmental conditions over the Central Amazon.
- 795 *Journal of Geophysical Research: Atmospheres, 122*(5), 2649-2666.
- 796 https://doi.org/10.1002/2016JD025993
- 797 Zipser, E. J., Cecil, D. J., Liu, C., Nesbitt, S. W., & Yorty, D. P. (2006). Where Are the Most
- 798 Intense Thunderstorms on Earth? Bulletin of the American Meteorological Society, 87(8), 1057-
- 799 1072. https://doi.org/10.1175/bams-87-8-1057

800 Figure 1. MODIS annual average land surface temperature (LST) in 2003-2019. (a) Daytime 801 LST. (b) Nighttime LST. The same color bar is used in top two panels. (c) Laplacian of daytime 802 LST. Only large positive values are shown to highlight the cooler river surface. (d) Laplacian of 803 nighttime LST. Only large negative values are shown to highlight the warmer river surface. 804 Figure 2. CMORPH annual average precipitation in 1998-2017 (a) throughout the day, (b) 805 between 1100 and 2300 LT, and (c) between 2300 and 1100 LT. The same color bar is used. 806 **Figure 3.** Diurnal and seasonal cycle of the river precipitation anomalies relative to the forest. 807 (a) Absolute anomalies of the entire river. (b) Percentage anomalies of the entire river. (c) 808 Percentage anomalies of the Negro. (d) Percentage anomalies of the Amazon. The hatching 809 denotes that more than 70% of the river grid cells agree in the sign of precipitation anomalies. 810 Panels (c) and (d) have the same color bar. 811 Figure 4. River precipitation percentage anomalies of each tributary. Average percentage 812 anomalies throughout the day are in blue; daytime average anomalies are in orange; and nighttime average anomalies are in purple. The box (quartiles) and whiskers (the minimum and 813 814 the maximum) show the spatial variation of the percentage anomalies in each tributary. The 815 tributaries are plotted in descending order of maximum channel width, except the UHE Balbina. 816 Figure 5. Afternoon CloudSat profiles (color contours) and average surface temperatures along 817 the swath (curve) near the Negro on November 6, 2006. The red line at the bottom denotes the 818 location of the river, and the shaded area is possibly raining/drizzling (CloudSat radar reflectivity 819 > -15 dbZ). 820 Figure 6. Linear relationship between the daytime river rainfall percentage anomalies and the 821 Laplacian of surface temperature. Climatological monthly means of six selected tributaries are

822	plotted ($n = 72$). The best linear fit derived from the ordinary least squares approach and its 99% confidence interval are also shown.
824	Figure 7. WRF annual average land surface temperature (LST) and Laplacians in July 2007-June
825	2010. (a) Daytime (1400 LT) LST. (b) Nighttime (0200 LT) LST. (c) Daytime (1400 LT)
826	$\nabla^2 \text{LST.}$ (d) Nighttime (0200 LT) $\nabla^2 \text{LST.}$
827	Figure 8. WRF annual average precipitation in July 2007-June 2010. (a) Average precipitation
828	1100-2300 LT. (b) Average precipitation 2300-1100 LT. (c) Average convective precipitation
829	1100-2300 LT.
830	Figure 9. Average horizontal moisture convergence profiles around 1100 LT before daytime
831	rainfall (a) above the river, and (b) above the surrounding forest. Data are categorized by total
832	daytime rainfall. Standard deviations are computed from bootstrap samples ($n = 1000$). The
833	horizontal lines and the grey shading indicate the mean planetary boundary top and its standard
834	deviation.
835	Figure 10. Decomposition of average horizontal moisture convergence profiles around 1100 LT
836	before daytime rainfall (a) above the river, and (b) above the surrounding forest. Standard
837	deviations are computed from bootstrap samples ($n = 1000$). The horizontal lines and the grey
838	shading indicate the mean planetary boundary top and its standard deviation.
839	Figure 11 . Changes in atmospheric humidity three hours ahead of strong afternoon rainfall (> 5
840	mm / 3 hr). Cases are categorized into shallow convection (SC) or no convection (NC) before the
841	rainfall occurs based on the maximum cloud fraction between 850 mb and 700 mb. Standard
842	deviations are computed from bootstrap samples ($n = 1000$).
843	Table 1.

844 Parameterization Schemes used in the WRF Simulation.

Parameterization	Scheme	Reference
Land	Noah land scheme	Chen & Dudhia, 2001
Radiation	Community Atmosphere Model 3.0 scheme	Collins et al., 2004

manuscript submitted to Journal of Geophysical Research: Atmospheres

Cumulus convection	Kain-Fritsch scheme	Kain, 2004
Planetary boundary layer	Yonsei University scheme	Hong et al., 2006
Cloud microphysics	WRF single-moment 5-class scheme	Hong et al., 2004